

Initiated by Deutsche Post Foundation

### DISCUSSION PAPER SERIES

IZA DP No. 16807

Has Intergenerational Progress Stalled? Income Growth over Five Generations of Americans

Kevin Corinth Jeff Larrimore

FEBRUARY 2024



Initiated by Deutsche Post Foundation

### DISCUSSION PAPER SERIES

IZA DP No. 16807

### Has Intergenerational Progress Stalled? Income Growth over Five Generations of Americans

Kevin Corinth American Enterprise Institute and IZA

Jeff Larrimore Federal Reserve Board

FEBRUARY 2024

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

### ABSTRACT

### Has Intergenerational Progress Stalled? Income Growth over Five Generations of Americans<sup>\*</sup>

We find that each of the past four generations of Americans was better off than the previous one, using a post-tax, post-transfer income measure constructed annually from 1963-2022 based on the Current Population Survey Annual Social and Economic Supplement. At age 36–40, Millennials had a real median household income that was 18 percent higher than that of the previous generation at the same age. This rate of intergenerational progress was slower than that experienced by the Silent Generation (34 percent) and Baby Boomers (27 percent), but similar to that experienced by Generation X (16 percent). Slower progress for Generation X and Millennials is due to their stalled growth in work hours—holding work hours constant, they experienced a greater intergenerational increase in real market income than Baby Boomers. Intergenerational progress for Millennials under age 30 has remained robust as well, although their income growth largely results from higher reliance on their parents. We also find that the higher educational costs incurred by younger generations is far outweighed by their lifetime income gains.

JEL Classification:	D31, E24, H24, J3, J62
Keywords:	full income, growth, generations, mobility, Millennials

### **Corresponding author:**

Kevin Corinth American Enterprise Institute 1789 Massachusetts Avenue, NW Washington, DC 20036 USA E-mail: kevin.c.corinth@gmail.com

<sup>\*</sup> The results and opinions expressed in this paper reflect the views of the authors and do not indicate concurrence by other members of the research staff or the Board of Governors. For helpful comments and discussion, we thank Richard Burkhauser, Jonathan Fisher, Maggie Jones, Bruce Meyer, Scott Winship, and participants at the 2023 National Tax Association conference.

### **1. Introduction**

A defining aspect of the American Dream is that the economic wellbeing of each generation should surpass that of the previous one. Whether this condition holds for younger generations has recently been called into question. A 2022 Gallup poll found that just 42 percent of Americans expect that today's youth will have a better life than their parents—down from 71 percent who felt that way in 1999 (Brenan 2022). Similarly, headlines in recent years have called Millennials (born from 1981–1996) the "unluckiest generation in U.S. history" (Van Dam 2020) and claimed that "many Millennials are worse off than their parents—a first in American history" (Luhby 2020).<sup>1</sup> Yet, when asked about their own financial situation compared to their parents at a similar age in the Federal Reserve's Survey of Household Economics and Decisionmaking (SHED), Millennial and Generation Z adults were nearly as likely as Baby Boomers to report doing better than their parents at the same age.<sup>2</sup> Hence, young adults appear to be more positive about their own financial progress than popular commentary suggests. Additionally, recent work by Twenge (2023) claimed that counter to recent narratives, intergenerational income growth for millennials in young adulthood far surpassed the intergenerational income growth for either Generation X or Baby Boomers. Gaining an accurate understanding of changes in economic wellbeing across generations, in light of these narratives, is important for assessing the state of the American Dream.

Previous research has provided important insights on intergenerational comparisons of wellbeing. Much of this research compares the economic wellbeing of adult children to their parents.<sup>3</sup> Studies of absolute mobility estimate the share of adults whose incomes exceed that of their parents at a constant age, often comparing contemporary adults to their parents without examining longer periods to document trends in absolute mobility (e.g., Urahn et al. 2012). One recent exception is Chetty et al. (2017). They combine cross-sectional survey data with individual tax records, allowing them to calculate trends in absolute mobility over longer time

<sup>&</sup>lt;sup>1</sup> That Millennials are the first generation to underperform their parent's generation is also the thesis of Filipovic (2020).

<sup>&</sup>lt;sup>2</sup> In the 2022 survey, 50 percent of Millennial adults and 51 percent of Generation Z adults felt they were better off financially than their parents at the same age. Among Generation X adults, 53 percent felt they were better off and among Baby Boomers, 56 percent did. In each of these generations an additional 21 to 26 percent said they were doing about the same as their parents financially (authors' calculations; Federal Reserve Board 2023).

<sup>&</sup>lt;sup>3</sup> Other research compares mobility in terms of other outcomes such as occupations (Long and Ferrie 2013; Song et al. 2020).

periods. When doing so, they find that the share of children who earn more than their parents fell from 92 percent among those born in 1940 to 50 percent among those born in 1984. Other studies focus on relative mobility, that is, the correlation between the position of an adult in the income distribution to the position of their parents in the income distribution at a constant age (Mayer and Lopoo 2005; Lee and Solon 2009; Chetty et al. 2014; Ward 2023).<sup>4</sup>

Although much of the previous literature is limited to comparing a relatively small number of generations (typically two) in a comparison of contemporary adults to their parents, whether the pace of intergenerational improvement has slowed could be as important as whether growth is positive. Additionally, a direct comparison of the economic wellbeing of adults to their own parents is not the only informative question about whether progress has slowed or reversed over generations. Current generations of adults might also compare themselves to the wider distribution of members of previous generations, not solely their own parents.<sup>5</sup> Finally, previous research has not evaluated changes in wellbeing using a comprehensive measure of income that adjusts for taxes, and includes both cash and in-kind transfers for the entire sample period to evaluate trends in a full array of resources.<sup>6</sup>

We address these questions by evaluating whether each generation as a whole has surpassed the previous one, comparing individuals from across seven generations—from the Lost Generation (born 1883–1900) to Generation Z (born 1997–2012)—using the Current Population Survey Annual Social Economic Supplement (CPS ASEC) from Flood et al. (2022).<sup>7</sup> We construct consistent comparisons by estimating for each generation at each age the distribution of a broad measure of income that accounts for tax liabilities and cash and in-kind transfers. The importance of assessing economic wellbeing using a broad measure of income has been

<sup>&</sup>lt;sup>4</sup> Survey-based studies have found mixed results. Chetty et al. (2014) use linked tax records to compare the income of adults of approximately 30 years of age in 2011 and 2012 to their family income as children, finding that a child whose parents are 10 percentile points higher up in the income distribution have on average a 3.4 percentile points higher income when they become adults.

<sup>&</sup>lt;sup>5</sup> Comparisons to one's own parents also differ from comparisons across generations in that one's parents are not always in the previous generation. As the age at which parents have their first child has increased (Mathews and Hamilton 2002), it produces more time for improvements from parent to child relative to the roughly constant length of societal generations.

<sup>&</sup>lt;sup>6</sup> We emphasize, however, that the cost of earning the income can change as well, as more young adults are investing in college and the cost of higher education has risen over time. We discuss implications of changes to human capital accumulation on results in section 5 of this paper.

<sup>&</sup>lt;sup>7</sup> While our data contain some members of these seven generations, we primarily focus on the five generations from the Greatest Generation (born 1901–1927) through Millennials (born 1981–1996) since we can observe them in mid-adulthood.

demonstrated by previous research that shows conclusions about trends in economic wellbeing over time are changed when using a full measure of income relative to a narrower one (see, for example, Piketty, Saez, and Zucman 2018; Larrimore et al. 2021; Burkhauser et al. Forthcoming; Auten and Splinter Forthcoming). Thus, using a broad measure of income could change conclusions about whether younger generations have outperformed older generations at a similar age. In addition, our 60-year sample period (1963–2022) allows for a longer period than most previous studies to put recent changes in historical context. Our analysis of incomes across generations can serve as a complement to recent research by Fisher and Johnson (2022) who used the Panel Study of Income Dynamics to consider income, consumption, and wealth across generations through adults born before 1985. It also complements Horpedahl (2021) who used the Federal Reserve's Distributional Financial Accounts to look at intergenerational wealth trends.

We find that at age 36–40 (our focal age range because it allows for comparisons across five generations) broad income has risen for every generation relative to the previous one, though at a slower rate for Generation X and Millennials. The median 36–40 year old from the Silent generation had a household income that was 34 percent higher than the Greatest Generation's median person for the same age range. Baby Boomers in this age range had a median income that was 27 percent above that of the Silent generation. The intergenerational income growth rate was a lower 16 percent for Generation X and 18 percent for Millennials. The rate of broad income growth was also positive at the 25<sup>th</sup> and 75<sup>th</sup> percentiles for each generation compared to the previous one, with a similar slowdown for the more recent generations.

The patterns of income growth over time have been affected by the growth in government transfers. In particular, the slowdown in intergenerational progress was more pronounced when looking at market incomes than when looking at broader income measures accounting for taxes and transfers. Nevertheless, Millennials in their late 30s still had market incomes that were 14 percent above that of Generation X.

The continued intergenerational progress we observe for those in their late 30s is also not due to changing rates of household formation. Changes in household formation boosted recent intergenerational progress in household incomes for those in their 20s, as more people now live with and rely on their parents in early adulthoods. This could reflect some financial distress

3

among young adults if they desire moving away from their parents but lack the financial resources to do so.<sup>8</sup> Yet, by age 31 less than 10 percent of Millennials lived within a household where their parents received over half the income. Consistent with these household formation trends, the improvements across generations for those in their late-30s are also apparent when looking at the incomes of individuals and couples, rather than households.

A major contributor to the slowdown in intergenerational progress for Generation X and Millennials is a slowdown in the growth of work hours. In their late 30s, the median members of the Silent Generation and Baby Boomers worked 27 to 32 percent more hours than the previous generation, while Generation X and Millennials worked on average 1 to 4 percent more than the previous generation. The work hours slowdown coincides with the stalling of female labor supply growth at the turn of the millennium, after which point Generation X and Millennials began to reach their late 30s. Holding work hours constant, we find that Generation X and Millennials experienced a larger intergenerational increase in market income than Baby Boomers and a somewhat smaller intergenerational increase than the Silent Generation.

Given the heightened concerns about student loan debt and the cost of college eroding this progress, we also consider the magnitude of intergenerational improvements relative to increases in the cost of higher education. Once accounting for growing financial aid, we find that the increase in the cost of higher education for Millennials relative to Generation X represents only three years of the increase in their household incomes. The burden of repaying this debt is frontloaded on early working years when earnings (for all generations) are lower, which do offset intergenerational gains for those who attend college as they enter adulthood. Nevertheless, when considered over the entire lifespan, the increased cost of college represents a small share of the increase in lifetime income for Millennials relative to previous generations.

Our results are consistent with previous research showing a general decline in absolute upward mobility over time (Chetty et al. 2017; Fisher and Johnson 2022). But in contrast to that work we find that for Millennials this decline stopped and income growth slightly increased. However, we also find that any acceleration in income growth is far smaller than that suggested by Twenge

<sup>&</sup>lt;sup>8</sup> Non-financial factors may also play a role, however, such as the rising age of first marriage (Julian 2021).

(2023), who observed that Millennials had about 9 times the intergenerational income growth at ages 25–34 than either Baby Boomers or Generation X.

The greater intergenerational progress that we observe relative to Fisher and Johnson (2022) in part reflects our broader income measure; additional years of data; and adjusting for inflation using the Personal Consumption Expenditures (PCE) price index, which like the Chained Consumer Price Index (Chained CPI) better reflects price changes than the traditional Consumer Price Index (CPI-U).<sup>9</sup> The greater consistency in intergenerational income growth that we observe relative to Twenge (2023) is likely due to our observing individuals across all years of the data.<sup>10</sup>

Given that we only observe the oldest Millennials at our comparison age in our dataset, our results are likely to somewhat understate the intergenerational progress of Millennials. This result counters some popular narratives that Millennials are worse off than previous generations, but it also provides optimism that the American Dream may no longer be fading. This is especially the case if rather than comparing themselves only to their parents, Americans define progress on the basis of how they compare to the full distribution of Americans in the prior generation, as we do in this paper.

The paper proceeds as follows: Section 2 describes our data, the construction of our broad income measure, and how we define and compare generations. Section 3 presents our main results on intergenerational progress. Section 4 documents the role of changes in work hours. Section 5 reports intergenerational progress by educational attainment. Section 6 considers racial differences in intergenerational progress. Section 7 reports the results of alternative specifications that account for incomplete observation of some generations and the inclusion of the value of health insurance as income. Section 8 discusses the results. Section 9 concludes.

<sup>&</sup>lt;sup>9</sup> The methodology of the CPI-U has changed over time. The Consumer Price Index retroactive series (CPI-U-RS) applies current methods to the historical CPI-U. Therefore, when comparing results in this paper to the CPI we use the CPI-U-RS rather than the CPI-U. For details on the CPI-U-RS, see Stewart and Reed (1999).

<sup>&</sup>lt;sup>10</sup> Although Twenge (2023) selects expansion years for the comparison, the notably different results for Generation X's intergenerational income growth that she finds in figure 5.15 comparing Gen X in 2004 to Baby Boomers in 1987 (at ages 25–34) and in figure 4.19 comparing Gen X in 2019 to Baby Boomers in 2004 (at ages 45–54) suggests the importance of the observation years for the trends.

### 2. Data and methodology

We use the CPS ASEC to construct our various income measures for each year from 1963 to 2022. The CPS ASEC is a nationally representative household survey used by the Census Bureau to estimate historical household income and poverty trends. It has the advantages of asking a large set of questions about income that elicit relatively more accurate responses than some other household surveys and being available annually for approximately six decades. We use the CPS ASEC to calculate for each individual or household two primary definitions of income—(i) market income and (ii) post-tax, post-transfer income, which starts with market income and subsequently adjusts for taxes and includes the value of cash and nonmedical in-kind transfers.<sup>11</sup>

Although the CPS ASEC is among the best sources of historical household income data, it lacks some of the necessary information to calculate these income measures because it lacks tax information prior to 2005 and the value of in-kind transfer benefits prior to 1979. Thus, we impute these missing income sources following the approach in Burkhauser et al. (Forthcoming). We impute federal income taxes, state income taxes, and payroll taxes for all years using NBER TAXSIM (Feenberg and Coutts 1993).<sup>12</sup> We use the imputed values of major nonmedical in-kind transfers—the Supplemental Nutritional Assistance Program (SNAP), rental housing assistance, and school lunch—from Burkhauser et al. (Forthcoming). These imputations rely on administrative data on aggregate caseload and spending data for each year, as well as predictions of recipients and benefit values on the basis of survey responses when such data were recorded in the CPS ASEC. We also correct for unemployment insurance underreporting in the CPS ASEC during the COVID-19 pandemic using tax-record based imputations from Larrimore, Mortenson, and Splinter (2023a). We similarly correct for differential nonresponse during the COVID-19 pandemic using the corrected weights from Rothbaum and Bee (2021). In Section 8, we discuss how other forms of income misreporting for which we do not correct may bias our results.

<sup>&</sup>lt;sup>11</sup> For income years 1963-1966, we exclude from market income all unearned income—both government income and interest, dividends, rent, and alimony—because all of these income sources are reported together as a single aggregate source. This will lead us to understate market income in 1963-1966, and because these non-government income sources are counted as market income in all later years starting in 1967, will lead us to overstate growth in market income for our earliest generations. Given that our focus is on relatively young working age adults (age 36-40), however, these unearned sources of income are likely small and thus the upward bias in market income growth is likely small. In 1967, these unearned government income sources represented just 2 percent of the total market income of 36 to 40 year old adults.

<sup>&</sup>lt;sup>12</sup> Because state tax laws prior to 1978 are not incorporated in TAXSIM, we apply state tax law as of 1978 to impute state income taxes for all years from 1963-1977, following the approach taken by Burkhauser et al. (Forthcoming).

Finally, while top-coding of income in the CPS ASEC leads to understatement of income at the top of the distribution, our focus on the median—and occasionally the 25<sup>th</sup> and 75<sup>th</sup> percentiles of the income distribution—instead of means allows us to avoid this source of bias.

We convert all nominal income values into 2019 dollars using the PCE price index. We use the PCE price index because unlike the CPI-U-RS it accounts for the ability of consumers to substitute across broad categories of items, and because unlike the Chained CPI-U, the PCE price index is available for our entire sample period.<sup>13</sup> In a sensitivity analysis, we also show major results using the CPI-U-RS instead. Notably, others have argued that even the PCE price index overstates inflation largely because it does not adequately account for new products and quality changes (Meyer and Sullivan 2012; Moulton 2018); this would suggest that our baseline estimates that adjust for inflation using the PCE price index could understate intergenerational progress.

We analyze two different sharing units. Our primary analysis takes the household as the sharing unit, reflecting all of the resources available to household members regardless of who brings the resources into the household. When using the household sharing unit, we equivalize resources using a square root equivalence scale. This reflects economies of scale in consumption, such that the cost of maintaining a constant standard of living when moving from one to two household members is equal to the cost when moving from two to four household members. In separate analyses we also consider the income of individuals and couples to distinguish between resources received by the members of a particular generation and the resources that result from changing child-bearing decisions. To the extent that adult children increasingly live with and rely on their parents to meet their material ends (so called "boomerang children"), this distinction could affect trends in household income growth. For the income of individuals and couples, we follow Piketty, Saez, and Zucman (2018), Larrimore, Mortenson, and Splinter (2022), and Auten and Splinter (Forthcoming), and allocate half of the couple's resources to each individual without

<sup>&</sup>lt;sup>13</sup> In 2023, the Census Bureau began using the Chained CPI-U to adjust incomes for inflation, rather than a version of the CPI-U that applies current methods historically (CPI-U-RS), recognizing that it is a better measure of actual consumer inflation experiences (Guzman and Kollar 2023). The Chained CPI-U is only available since 2000, so cannot be used in earlier years. Because the PCE price index and Chained CPI-U find similar rates of inflation, and the PCE price index is available from the beginning of our series, we use the PCE price index for all years.

further adjustments for economies of scale.<sup>14</sup> We allocate all market income such as earnings to the individual who earns it, and we allocate taxes equally within couples who are expected to jointly file a tax return. For cash transfer income captured in the CPS ASEC, we allocate the transfer income based on the recipient recorded in the survey. For imputed in-kind transfer income, we allocate transfers to the householder identified in the CPS ASEC.

We group individuals into generations based on the year in which they were born: the Lost Generation (1883–1900), the Greatest Generation (1901–1927), the Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996), and Generation Z (1997–2012). These classifications, taken from those created by Pew Research (Dimock 2019), have the advantage of being used in previous research, reports, and public commentary.<sup>15</sup> Each of the generation classifications since the Silent Generation also spans a relatively similar amount of time, from 15 to 18 years, allowing for consistent comparisons of income growth across generations. We compare generations by examining the distribution of income across all individuals in a given generation of a given age (or age range). Because members in a given generation were born over a 15- to 18-year period, a given generation will be of a given age in different calendar years. For example, members of Generation X were 35 years old between 2000 (those born in 1965) and 2015 (those born in 1980).

Our data sample includes calendar years 1963 through 2022, and so we can only make comparisons of generations which reached a given age in some year during that range.<sup>16</sup> Figure 1 indicates the calendar years in which a generation's members were of a given age. The dashed red lines at 1963 and 2022 indicate the earliest and latest years of our sample. Thus, we only observe a generation's members at a given age if it is at least partially contained by the dashed red lines. For example, at age 20, we observe three years of the Silent Generation (those born in 1943–1945), as well as all Baby Boomers, Generation X and Millennials. We do not observe the Greatest Generation at age 20 because they were born prior to 1943. Similarly, we do not

<sup>&</sup>lt;sup>14</sup> Subsequent references in this paper to couple incomes always refer to the income of the individual alone if single, and the equal-split income of the individual and their spouse if married.

<sup>&</sup>lt;sup>15</sup> Dimock (2019) does not provide a beginning year for the Greatest Generation or the span of the Lost Generation. We define the Lost Generation as those born between 1883 and 1900 based on Strauss and Howe (1991), who also define the Greatest Generation as starting with those born in 1901.

<sup>&</sup>lt;sup>16</sup> To ensure that results are not driven by the pandemic experiences, we also conducted our analyses ending in 2019 and produced similar results.

observe those members of Generation Z who were born after 2002 at age 20 or later in adulthood.

For our purposes, we are concerned about observing adults at a prime working age after educational investments have been made. This makes ages 36–40 of interest, because these encompass prime working ages for which we can observe five generations—the latest born members of the Greatest Generation, all of the Silent Generation, Boomers and Generation X, and earlier born Millennials. Because we only observe the youngest members of the Greatest Generation, we are likely to overstate the incomes of all Greatest Generation members (most of whom reached age 36–40 before our data sample begins in 1963) and thus understate income growth from the Greatest Generation to the Silent Generation. And because we only observe the oldest Millennials, we are likely to understate the incomes of all Millennials (most of whom will reach age 36–40 after our data sample ends in 2022) and thus understate income growth from Generation X to Millennials. In Section 7 we show the robustness of our results when adjusting for our incomplete coverage of Millennials at age 36–40 in our sample.

### 3. Intergenerational Progress

We begin by comparing income growth across five generations at our focal age range of 36–40 years old. Table 1 reports the percent increase in income for each generation relative to the immediately preceding generation. We use the couple and the household sharing units. Panel A reports growth in market (i.e., pre-tax, pre-transfer) income, whereas Panel B reports growth in post-tax, post-transfer income.

Focusing first on market income in Panel A, there are two notable takeaways that apply for both the couple and household sharing units. The first is that intergenerational progress has clearly slowed since the Baby Boom generation, although it remains positive. But second, despite the perception that slowing intergenerational progress is a recent phenomenon, the substantial slowdown did not start with Millennials. It began a generation earlier with Generation X. Intergenerational progress actually accelerated slightly for Millennials, although certainly not approaching the pace observed for the Baby Boom generation.

Looking at the patterns for median household market income by generation, the median income of Baby Boomers in their late-30s was 31 percent above that for similarly aged adults in the

9

Silent Generation. Progress slowed substantially for Generation X—their incomes increased by 10 percent relative to Baby Boomers—and then ticked up slightly for Millennials whose incomes rose by 14 percent relative to Generation X. This slowdown in household incomes across generations is more pronounced than that seen for couple-level incomes, which reflects the slowing of the decline in household size. A substantial decline in household size for Baby Boomers served to increase the growth rate of their household equivalized income, whereas a negligible decline in household size for Generation X and Millennials did little to boost their growth rates.<sup>17</sup>

Although market income is an important indicator of progress, it does not reflect the full set of resources that individuals have available for consumption.<sup>18</sup> In Panel B, we therefore expand the income definition to a post-fiscal measure, after taxes and transfers (excluding health insurance).

Consistent with the increase in transfers and the decrease in tax rates for most adults over the past several decades, the slowdown in intergenerational progress is softened when using these broader measures. The growth rate for post-tax, post-transfer median income was similar for the Millennial, Generation X, and Baby Boom generations if looking at couple level incomes. For each of the three generations, incomes of people in their late-30s increased between 15 and 19 percent relative to similarly aged adults in the prior generation. Again reflecting changes in household size, there is a greater slowdown in household income growth than was observed when looking at the equal-split incomes of couples.

It is possible that the patterns of intergenerational progress depend on where in the income distribution we focus. Hence, in Table 2 we consider the progress for those lower in the distribution at the 25<sup>th</sup> percentile and in Table 3 we consider the 75<sup>th</sup> percentile. In the Appendix (Appendix Tables A1 and A2) we show the 10<sup>th</sup> and 90<sup>th</sup> percentiles.

<sup>&</sup>lt;sup>17</sup> Baby Boomers aged 36-40 experienced an 11.4 percent decline in average household size relative to Silent Generation members of the same age, Generation X members aged 36-40 experienced a 0.3 percent decline in average household size relative to Baby Boomers of the same age, and Millennials aged 36-40 experienced a 1.1 percent decline in average household size relative to Generation X members of the same age, based on an arithmetic average of household sizes over all years in which a given generation was aged 36-40 (Authors' calculations; Census Bureau 2022).

<sup>&</sup>lt;sup>18</sup> Market income should also not be interpreted as the incomes that individuals would receive in the absence of government interventions. Because individuals change their labor market decisions and businesses adjust their practices in response to incentives, people's market incomes may be more or less that what we observe if these programs did not exist. Additionally, some government interventions, such as minimum wages, affect individuals' market earnings directly.

The broad conclusions are similar at these percentiles to those for the median—intergenerational progress has generally remained positive even though it has slowed relative to earlier generations. Yet, these findings are indicative of inequality trends over the past half-century. Looking at Generation X, specifically, those at the 25<sup>th</sup> percentile of the distribution had almost no intergenerational progress in market incomes (at either the household or couple sharing unit level) relative to Baby Boomers whereas those at the 75<sup>th</sup> percentile saw growth in market incomes of around 16 to 17 percent. Nevertheless, once accounting for taxes and transfers, we again see growth in 25<sup>th</sup> percentile incomes for those in Generation X. For Millennials, income growth accelerated at the 25<sup>th</sup> percentile, while modestly continuing its slowdown at the 75<sup>th</sup> percentile, for each income and sharing unit definition. This acceleration for Millennials at the 25<sup>th</sup> percentile was particularly pronounced for market income—at the household sharing unit level, market income grew by 13 percent for Millennials compared to just 1 percent for Generation X.

We can also compare generations across all ages, rather than solely the focal age range of 36–40. Figure 2 (market income) and Figure 3 (post-tax, post-transfer income) show the median income of each generation at each individual age.

Looking first at the incomes of individuals and couples (Panel A of Figures 2 and 3), we see that the patterns observed for those aged 36–40 described above are broadly consistent throughout the observable life-course. Among those in their late 20s or older, each generation is doing better than the previous generation, although improvements are more modest and came slightly later for Generation X and Millennials than was the case for earlier generations. This is true using each of the two income definitions. It will, however, be important to continue to monitor the age-income profile of Millennials in future years to see if the income growth they began to establish in their mid-30s is maintained as they continue to age.

In Panel B of Figures 2 and 3, we switch to using a household income definition. At the household level, intergenerational improvements in income are even clearer at all age ranges, including young adulthood. This is consistent with individuals increasingly living with and relying on their parents well into their 20s, although improvements from relying on parental resources do not reflect the same type of financial progress as improvements from one's own

11

income.<sup>19</sup> As shown in Appendix Figure A1, the share of each generation who were dependent on their parents in their 20s and 30s—defined as living in a household in which their parents or their spouse's parents received over half of the household's income—has risen with each generation. The share of a generation's members who were dependent on their parents fell below 10 percent for the first time by age 26 for Baby Boomers, by age 28 for Generation X, and by age 31 for Millennials. Despite their delay in achieving independence from their parents, Millennials saw their household incomes continue to rise well into their 30s. This indicates that Millennials were able to achieve a higher financial standard of living than previous generations by sharing resources in their 20s (perhaps while continuing their education) and then continued to maintain this higher standard of living when setting off on their own in their 30s.

### 4. Changes in Work Hours

Work behavior has changed dramatically over the time period we study (1963–2022), especially due to the rise in female labor force participation until the turn of the millennium. Differences in intergenerational growth rates of work hours—on the extensive and intensive margin—by individuals and their spouses could affect the income growth estimates we reported in the previous section.

Figure 4 reports the labor force participation rate of prime age adults (aged 25 to 54) from 1963 to 2022. The dashed black lines indicate the mean labor force participation rate of all prime age adults (not only the adults who were actually members of the generation) during the time period in which each generation had members aged 36–40. Prime age labor force participation rose from 70 percent when the Greatest Generation was observed in their late 30s, to 75 percent when the Silent Generation was in their late 30s, and further to 83 percent when Baby Boomers were in their late 30s. Prime age labor force participation then decreased slightly to 82 percent when Generation X and Millennials were in their late 30s. This figure makes clear that the structural labor market trends over the past six decades served to boost work hours for the Silent Generation and Baby Boomers, before completely stalling for Generation X and Millennials.

<sup>&</sup>lt;sup>19</sup> According to the 2019 SHED, nearly half of adults ages 22 to 24 lived with a parent, as did over one-fourth of 25 to 29 year olds (Canilang et al. 2020).

In Table 4, we report the hours worked by the median and mean member of each generation between the ages of 36 and 40. We add the hours worked among both partners in a couple, and we do not condition on work participation for inclusion in the sample.<sup>20</sup> At the median, the Silent Generation worked 27 percent more hours and Baby Boomers worked 32 percent more hours than the previous generation. Growth in work hours almost completely stalls for the younger generations, with Generation X working 1 percent more hours and Millennials working 4 percent more hours than the previous generation.<sup>21</sup> Intergenerational growth rates in mean hours portray a consistent picture, with mean work hours growing by 14 percent for both the Silent Generation and Baby Boomers, then falling to -2 percent for Generation X and 0 percent for Millennials.<sup>22</sup>

We next quantify the extent to which the higher growth rates in work hours by the Silent Generation and Baby Boomers explains their higher rates of intergenerational income growth. We do so by calculating how market income would have grown for each generation had their work hours not changed relative to the previous generation. We focus on market income because taxes and transfers are directly linked to work through earnings, making it difficult to determine how taxes and transfers would change if work hours had counterfactually matched the work hours of the previous generation. In order to hold work hours constant, we retain our focus on 36-40 year old members of each generation, calculate the mean hours worked by couples in the middle quintile of the market income distribution, and assume that couples in the middle quintile of the following generation worked the same number of hours on average—but at their actual hourly wage—for purposes of making pairwise comparisons. Thus, we calculate the percent change in mean market income of generation t, when holding work hours constant, as

<sup>&</sup>lt;sup>20</sup> This approach does not affect the mean hours worked relative to calculating hours at the individual level, except when members of a couple differ in age. It can affect median hours, particularly as labor force participation rates change with the growth in two-earner couples.

<sup>&</sup>lt;sup>21</sup> Appendix Figure A2 shows the mean hours of work for members of each generation for their prime working years (ages 25–54). These results are broadly consistent with Table 4, except that Generation X slightly exceeded the hours of work seen among Baby Boomers and Millennials in their late 20s before work hours equalized across the generations in their mid-30s.

 $<sup>^{22}</sup>$  We focus on hours worked here, rather than the implied hourly wage since we do not observe hourly wages for those who are not working and can therefore only consider hourly wages among workers. However, the trend in wages among workers tells a consistent story to what we present here, as seen in Appendix Table A3. Among those in couples where at least one person works, the intergenerational growth in mean and median hourly wages (calculated as the couple's labor earnings divided by hours worked) was slower for the Baby Boom generation than for either Generation X or the Millennial generation.

$$\%\Delta m_t = \frac{w_t h_{t-1} + n_t}{w_{t-1} h_{t-1} + n_{t-1}} - 1 \tag{1}$$

where  $m_t$  is the mean market income,  $w_t$  is the mean hourly wage (weighted based on actual hours worked by each couple in the generation),  $h_t$  is mean work hours, and  $n_t$  is the mean nonlabor income of generation t.<sup>23</sup> Note that the numerator multiplies the weighted average hourly wage rate of generation t by the mean work hours of generation t - 1, allowing us to determine the mean market income of generation t if their mean hours worked had not changed relative to the previous generation. The implicit assumption is that the hours worked by each couple in the younger generation should be adjusted by the same proportion in this counterfactual scenario.<sup>24</sup>

Table 5 reports the extent to which changes in work hours affect intergenerational growth in the market income of couples, focusing on the mean of the middle quintile of each generation at age 36–40. The first row of Table 5 reports the actual intergenerational increases in market income, which differ only slightly from the median-based values reported in Table 1. The second row of Table 5 reports the percent increase in market income of each generation holding work hours constant, counterfactually assuming their mean work hours equaled the mean work hours of the previous generation as described by Equation (1). Holding work hours constant, intergenerational progress slowed from 14 percent for the Silent Generation to just 2 percent for Baby Boomers, before speeding up to 7 percent for Generation X and 10 percent for Millennials. The third row of Table 5 reports the percent increase in market income due to the change in work hours, i.e., the difference between the actual increase in market income and the increase in market income holding work hours constant. The rise in work hours did the most to boost the market incomes of the silent Generation and Baby Boomers, leading their market incomes to rise by 18 percent and 19 percent respectively, while only boosting the market income growth of Generation X and Millennials by about 1 percent each. The rise in work hours explains 56 percent of the market income growth for the Silent Generation and 90 percent for Baby Boomers, while explaining only 16 percent for Generation X and 6 percent for Millennials.

<sup>&</sup>lt;sup>23</sup> We calculate the implied hourly wage for each individual or couple as their observed labor income divided by their observed work hours. Note that labor earnings represent at least 97 percent of market earnings for those ages 36 to 40 in each generation.

<sup>&</sup>lt;sup>24</sup> See Appendix B where we show that (i) mean market income can be written as shown by the numerator in the first term of Equation (1), and (ii) Equation (1) implies that hours worked are adjusted proportionally for each couple, by the ratio of mean work hours of the older generation to mean work hours of the younger generation.

### 5. Intergenerational Progress by Education

Income growth across generations is also a reflection of the rising level of human capital investment. At our focal ages of 36–40, nearly half (47 percent) of Baby Boomers had a high school degree or less, whereas 27 percent had at least a bachelor's degree (Table 6). Among Millennials, just 33 percent had a high school degree or less while 43 percent had at least a bachelor's degree. Additionally, the cost of college has risen substantially in recent decades (National Center for Education Statistics 2022), increasing the cost of these educational investments. It is, therefore, possible that while incomes for Millennials in their late-30s are higher than that for any previous generation, the costs of their educational investments outweigh these gains.

We first explore this possibility by considering the intergenerational income growth for those with the same level of educational attainment. These results can be seen in Table 7 and Figure 5. For those with an associate degree or less in their late-30s, incomes did rise from one generation to the next although the gains were relatively small for Generation X. Additionally, we note that when looking at the entire lifespan in Figure 5, members of Generation X in their 40s with a high school degree or less have lower incomes than their Baby Boomer counterparts of the same age and the same educational attainment.

Among those with at least a bachelor's degree, rates of income growth across generations are higher, although growth has still slowed.<sup>25</sup> Millennials with a bachelor's degree, but no graduate education, earned 10 percent more than similarly educated members of Generation X in their late-30s. Yet this is far less income progress than the 17 percent growth in incomes that Baby Boomers with a bachelor's degree experienced relative to the Silent generation.

In dollar terms, Millennials aged 36–40 with a bachelor's degree had equal-split couple incomes that were \$4,600 higher than did equivalently aged members of Generation X.<sup>26</sup> Members of

<sup>&</sup>lt;sup>25</sup> Compositional changes in who attends college will, admittedly, affect these comparisons. As more students attend college, this can decrease the earnings growth of the "high school degree or less" group, by shifting higher earners into the "some college or more" groups. But if these additional enrollees also have lower earnings potential than other college attendees, it can also slow the growth of the Bachelor's degree or more group.

<sup>&</sup>lt;sup>26</sup> Couple income is used here to focus on the individual's own earnings rather than support from others in the household. Additionally, since we use an equal split for couple income, the dollar values can be directly compared to the costs of education, which cannot be done with size-adjusted incomes. When using size-adjusted household

Generation X with a bachelor's degree had size-adjusted household incomes that were \$6,700 above that of Baby Boomers. Zooming out, over the 33-year period for each individual from age 25 to age 57—the oldest age at which we can observe members of Generation X—the median size-adjusted household-incomes of Generation X with a bachelor's degree exceeded that of Baby Boomers with a Bachelor's degree by \$198,000.

In 1975–1976, when a Baby Boomer born in the middle of the generation turned 21, the average tuition, fees, room, and board in the United States cost \$7,824 (in 2019 dollars), as reported in Table 8. This compares to an average cost of \$10,811 when a member of Generation X in the middle of their generation turned 21. Hence, for a four-year education, someone in the middle of the Generation X cohort would spend approximately \$12,000 more than someone in the middle of the Baby Boomer generation. This represents just 6 percent of the \$198,000 33-year income gain for Generation X over Baby Boomers through age 57. While we do not yet observe Millennials above the age of 41 in our dataset to have a sense of their more complete lifetime incomes, the higher annual incomes of Millennials suggests that their income gains over Generation X will also far outweigh their higher educational investments. The four-year educational cost for Millennials exceeds that of Generation X by about \$38,000. This is equivalent to about eight times the \$4,600 one-year income gap between Millennials and Generation X with a Bachelor's degree in their late-30s. Additionally, despite having many working years left, between ages 25 and 41 alone the median income of Millennials exceeded that for Generation X by just over \$60,000-more than 50 percent above the increase in education costs between these generations.

Nevertheless, young adults may also not feel the benefits of these gains early in their career, due to the timing of student loan repayments. A \$38,000 student loan at 6% interest with a typical 10-year repayment term would require annual payments of \$5,000.<sup>27</sup> Looking at Figure 5, from ages 25–29 the intergenerational couple-level income gains were smaller for those with a

incomes, the results are qualitatively similar except that Millennials in their 20s had greater improvements over previous generations that are more likely to exceed increases in education costs.

<sup>&</sup>lt;sup>27</sup> We use \$38,000 here to reflect the increase in the sticker price of college relative to prior generations. However, student loan debt is generally far lower. Based on the Survey of Consumer Finances, the median student loan debt for those with outstanding balances in 2019 when a person born in the middle of the Millennial generation turned 30 was \$25,500. This is up from \$13,300 in 2001 when a person born in the middle of Generation X turned 30 (Board of Governors of the Federal Reserve System 2023; Authors' calculations).

Bachelor's degree—just \$2,600 per year for Millennials relative to Generation X. This can contribute to the feeling of generational stagnation for those with student loans, even if there is likely to be progress over their lifetime after the completion of student loan repayments.

We also note that this comparison of sticker price of college over time likely overstates the challenge of making up the increased educational cost. This is because the rise in financial aid over this period mitigated the rising sticker price of college. Between 1990 and 2010, real annual financial aid per full time equivalent undergraduate student increased by almost \$11,000, with about \$6,500 of that increase due to grants rather than loans and work-study programs and which over a four-year period would total about \$26,000, over two thirds of the college cost differential for Millennials versus Generation X (Dynarski, Page, and Scott-Clayton 2022; authors' calculations). Hence, while the increase in educational costs could partially offset the intergenerational income gains and fully offset them if only looking at income early in adulthood, the higher cost reflects a relatively small share of the lifetime intergenerational progress for those who complete a degree.

### 6. Intergenerational Progress by Race

Intergenerational progress may also look different across races and ethnicities given the unique challenges and opportunities faced depending on one's race and ethnicity. Changes to the CPS ASEC race and ethnicity classifications preclude a full analysis of racial and ethnic trends in intergenerational progress since we use data back to the early 1960s. While we would prefer to be able to also examine other groups such as Asians, Native Americans, and those of Hispanic ethnicity, we are at least able to compare the intergenerational progress for Black Americans to nonblack Americans over our entire time period.

Consistent with that seen for the population as a whole, intergenerational income progress in their late-30s has been positive for both Black and nonblack adults for each of the past four generations (Table 9 and Figure 6). However, we also observe that there has been a greater slowdown in intergenerational progress for Black Americans than is seen for nonblack Americans. Black Baby Boomers saw household (post-tax, post-transfer) incomes that were 36 percent above Black members of the Silent generation at the same age. The intergenerational income growth for Black Millennials was 16 percentage points lower (20 percent growth across

17

generations). Among nonblack adults, there was a smaller 9 percentage point slowdown in household income growth rates between these two generations (27 percent income growth for Baby Boomers and 18 percent for Millennials). Consequently, while intergenerational income growth was far faster for Black adults in the Silent, Baby Boom, and X generations than was seen among non-Black adults in those generations, for the Millennial generation the intergenerational growth rates across racial groups are far more similar.

Given the faster intergenerational income growth for earlier generations of Black adults than nonblack adults, the gap in household incomes has declined across generations. Black members of the Greatest generation had size-adjusted household incomes that were 46 percent below that of nonblack members of that generation in their late 30s. In the Silent Generation the gap was 69 percent, in the Baby Boom generation the gap was 26 percent, and in the Millennial generation it was 23 percent. Yet the convergence of intergenerational income growth rates has meant that the progress at closing this racial income growth has slowed for the most recent generations. Between Generation X and the Millennial generation there was just a 1 percentage point reduction in this racial income gap.

### 7. Robustness of Results to Alternate Specifications

#### Incomplete generations

One limitation of the approach above is that the youngest generations are incomplete. In particular, the analysis of those aged 36–40 only captures the oldest Millennials. For example, the youngest Millennials (born in 1996) do not turn 40 until 2036 and thus are not observed in the sample. Because we only observe the earliest born Millennials, income growth may be biased when comparing Millennials to Generation X. It can also affect intergenerational progress in the earliest generations, depending on how the oldest members of each generation fared compared to younger members of the same generation. We therefore calculate income growth between each generation looking only at those members who were born in the first three years of the generation. This comparison is provided in Table 10.

When only including the oldest members of each cohort, intergenerational progress has shown a more pronounced slowdown. However, this is attributable to faster growth in earnings for the oldest Baby Boomers relative to the oldest members of the Silent generation than were observed

for those cohorts as a whole in Table 1. Among Millennials and Generation X, these results are consistent with those previously observed for the complete cohorts.

We further explore the effect of incomplete generations on our results by splitting each generation in half, a younger half and an older half. This mitigates the bias from observing partial generations because, compared to a full generation, the unobserved members of a half generation will be closer in age to the observed members. Analyzing half-generations also provides a more granular assessment of progress over time by smaller cohorts of Americans. In Figure 7 we subdivide each generation into those born in the first half of the generation and those born in the second half. Comparing the oldest half of each generation to each other (solid lines) we again see that each generation consistently improved over those at a similar age range of the generation before. The same is true when comparing the younger half of each generation (dashed lines) to the one before. As expected, however, since there is less time for progress when considering half-generations, if we look at each half-generation relative to the half-generation before (such as comparing older Generation X members to younger Baby Boomers), the growth is smaller with occasional ages where a generation is worse off than those a half-generation earlier.

#### Effects of including health insurance benefits with income

Thus far we have excluded health insurance from our income measure, due to substantial disagreement over whether and how to include it. Here we consider the implications of the inclusion of health insurance by adopting the full income measure from Burkhauser et al. (Forthcoming) that is the same post-tax, post-transfer income concept shown previously except that we add the full ex-ante market value of health insurance. Because the CPS lacks information on health insurance received outside of 1979-2014, we follow the imputation procedures from Burkhauser et al. (Forthcoming) to impute the market value of public health insurance and employer sponsored coverage in the years when it is unavailable. We also limit the analysis to years before 2020 due to a lack of health insurance data for 2020 through 2022.<sup>28</sup>

<sup>&</sup>lt;sup>28</sup> For this analysis we also limit the focal ages to 36–38, which is the oldest age at which Millennials could be observed in 2019. To allow for a direct comparison of how health insurance affects results, rather than the change in years, in Table 8 we also show the post-tax, post-transfer income growth excluding health insurance—matching Panel B of Table 1 except for the change in end-year and age range.

For this analysis, we set the value of health insurance at its full ex-ante market value, which for public health insurance (Medicare and Medicaid) we take as the average cost of the government to provide coverage, and for employer sponsored coverage we take as the employer paid premium. For public health insurance, we calculate the average cost across separate risk classes based on age and disability status, determined separately for each state. Using the market value of health insurance captures the full value of insurance provided without suffering from the perverse consequence from instead relying on medical expenditures made on behalf of an individual, which would lead those who get sick and thus require more medical care to appear as having higher incomes. The market value approach is used by the Congressional Budget Office (CBO) for their household income distribution estimates (Congressional Budget Office 2012) and has been frequently adopted in recent academic research (Burkhauser, Larrimore, and Simon 2012; Meyer and Sullivan 2012; Armour, Burkhauser, and Larrimore 2013; Kaestner and Lubotsky 2016; Elwell, Corinth, and Burkhauser 2020; Larrimore et al. 2021).

When including the market value of health insurance (Panel B of Table 11, and Figure 8), intergenerational income growth remains uniformly positive. However, the inclusion of health insurance somewhat strengthens income growth for Generation X and weakens it for Millennials.

To understand the faster intergenerational income growth for Generation X and slightly slower growth for Millennials when including health insurance, it is helpful to consider the long-run trajectory of health insurance spending. Based on data from the Centers for Medicare and Medicaid Services (2023), national health expenditures per capita increased by 3.5 percent per year (in real terms using the PCE price index) from 1990, when someone born in the middle of the Baby Boom turned 35, to 2008, when someone born in the middle of Generation X turned 35. From 2008 through 2019, health expenditures per capita increased by a more modest 2.0 percent per year. Consequently, including the health insurance benefits that are paid by someone else (either an employer or the government) more rapidly increased the observed resources for Generation X than was the case for Millennials.<sup>29</sup>

<sup>&</sup>lt;sup>29</sup> Increases in expenditures may result from medical advances, which result in real improvements in standards of living, or from cost increases, which do not. The increased costs of medical care will be included in the PCE price index, although the difficulty in separating these two components of the rise in medical expenses are an additional reason why we opt to include health insurance separately from other in-kind benefits.

#### Alternate inflation correction

In the previous sections we used the PCE price index to adjust for inflation. This is because the PCE price index is available historically and, like the Chained-CPI that is used by the Census Bureau for adjusting income series (Guzman and Kollar 2023), it captures changes in consumer purchasing decisions better than the CPI-U-RS. However, much of the prior research has used the CPI-U-RS as the inflation measure since this was used by the Census Bureau for their inflation adjustments until 2023. We therefore also consider how the choice of inflation measure affects our results.

Table 12 shows the intergenerational growth in median income using the CPI-U-RS to adjust for inflation alongside the results using the PCE price index (panels C and D). Since our earlier results used the PCE price index as the inflation measure, Panels C and D perfectly match Table 1 but are reproduced here to aid the comparison.

Because the CPI-U-RS is known as an upper-bound on inflation (U.S. Bureau of Labor Statistics 2019), using this measure to adjust for inflation causes intergenerational income growth to appear slower for each generation. In particular, it reduces the apparent market income growth for Generation X by at least half—from 8 percent growth for couples using the PCE to 2 percent using CPI and from 10 percent at the household level using PCE to 4 percent using CPI. Hence, the Generation X slowdown is more substantial when using CPI inflation, although growth did remain at least slightly positive for each of our primary income measures. We also continue to see a rebound in income growth from Generation X to Millennials when using CPI inflation.

Figure 9 shows the full age-income profiles for market income and post-tax, post-transfer income using the CPI-U-RS inflation measure. Consistent with that seen in Table 12, the growth across the age distribution is muted relative to that seen when using PCE inflation. Additionally, when looking at market income for couples, there was very little income growth from the Baby Boom Generation to Generation X, and growth for the Millennial generation only emerges starting in their mid-30s. Hence, while it is not accurate to say that the Millennial generation is doing worse than prior generations—at least with respect to annual income—when using the narrower sharing unit and especially when using market incomes the improvement is clearly muted when using CPI to adjust for inflation until people reached their mid-30s.

21

### 8. Discussion

We find that each of the past four generations of Americans was better off than the previous one. Millennials—the most recent generation we can assess in their late-30s—had a median household income at age 36–40 that was 18 percent higher than that of the previous generation at the same age. This rate of intergenerational progress was slower than that experienced for the Silent Generation (34 percent) and Baby Boomers (27 percent), but slightly faster than that experienced by Generation X (16 percent). The slower progress for younger generations is a result of their stalled growth in hours worked. Most of the intergenerational market income gains for the Silent Generation and Baby Boomers resulted from the fact that they worked substantially more hours than the previous generation, working 27 to 32 percent more hours at the median. Holding work hours fixed in each intergenerational comparison, Millennials have experienced a larger intergenerational market income gain than Generation X and Baby Boomers. Moreover, the Millennial growth rate will likely increase further as time passes, because we only capture the oldest Millennials in our comparisons and there will continue to be Millennials aged 36–40 until the year 2034 during which time incomes are likely to grow further.<sup>30</sup>

These optimistic results contradict a perception that Millennials are falling behind previous generations. One possible explanation relates to variation in growth rate trends across the income distribution. Household income among Millennials has grown at each part of the distribution, but the rate of growth has slowed for higher income Millennials while it has sped up for lower income Millennials. For the 25<sup>th</sup> percentile 36–40 year old, income growth relative to the previous generation was 69 percent (8 percentage points) faster for Millennials than Generation X, while for the 75<sup>th</sup> percentile 36-40 year old, growth was 6 percent (1 percentage point) slower for Millennials than Generation X. Higher income Millennials who compare themselves to higher income members of Generation X would accurately sense that intergenerational progress is slowing down for their socioeconomic group even though the growth rate remained positive.<sup>31</sup>

<sup>&</sup>lt;sup>30</sup> The Congressional Budget Office projects that that real per capita personal income will grow by 17 percent over the next decade, 2023-2033 (Congressional Budget Office 2023; Authors' calculations).

<sup>&</sup>lt;sup>31</sup> A related concept of the appropriate comparison group is trends by gender. As discussed by Twenge (2023), while women's earnings have risen across generations men's earnings have not. This can contribute to feelings of stagnation for men who are comparing themselves to men of earlier generations. When looking at couple-level and household-level incomes, intergenerational income growth is similar for men and women in their mid-30s (Appendix Table A4) since a majority of people in all generations are married at these ages, although changes in individual earnings power can still affect the feelings of stagnation.

This could especially be the case if they focus on salient markers of affordability like rising home prices and rents relative to incomes in expensive coastal cities. At the same time, our use of a general inflation measure accounts for rising prices of all items for the average consumer, so although the prices of some items have risen faster than overall inflation they are counterbalanced by the prices of other items whose price has grown more slowly.<sup>32</sup>

Another possible explanation for the perception that Millennials are falling behind is the increasing length of time they spend investing in their human capital and relying on others for support until reaching their 30s. Millennials in their late 30s were 7 percentage points more likely to have earned at least a Bachelor's degree than Generation X, and 4 percentage points more likely to earn a graduate degree. Perhaps as a result, Millennials' market income received only by themselves or their spouses is no higher than that of Generation X until around age 30, at which point Millennials begin to pull away. Additionally, student loan repayments in their 20s will offset some or all of the income gains for those who borrowed to attend college. By contrast, Generation X began to pull away from Baby Boomers around their mid-20s, and Baby Boomers began to pull away from the Silent Generation in their early 20s. Despite the longer time it takes for Millennials to surpass the previous generation in terms of their own market income, their post-tax, post-transfer income of their entire household (including the parents of "boomerang" children) exceeds that of Generation X from the beginning of adulthood. Yet most of the income boost in their 20s is a result of living in households with higher income individuals (i.e., their parents) rather than the effects of market income or transfers received by Millennials themselves. That Millennials' generational progress throughout their 20s is a result of being supported by others could contribute to a perception of stagnation.

As Millennials enter their 30s and 40s, we can rule out the possibility that paying back their higher student loans accumulated via increased educational investment will be large enough to outweigh their income gains at the median of the distribution. When comparing the intergenerational gains for those with a Bachelor's degree to the change in the cost of college, the rising educational expenditures are relatively small when compared to their increases in income. Millennials in their late-30s with a bachelor's degree had annual couple-level incomes

<sup>&</sup>lt;sup>32</sup> Moreover, while the share of income spent on housing has risen over time, the share of disposable personal income spent on food in the United States fell from 17 percent in 1960 to 11 percent in 2022 (U.S. Department of Agriculture 2023).

that were \$4,600 higher than Generation X and \$11,300 higher than Baby Boomers. Before accounting for the increased generosity of financial aid, Millennials faced an average four-year college cost that was \$38,000 higher than Generation X—representing about 8 years of Millennials' \$4,600 higher annual mid-career income—and a college cost that was \$50,000 higher than Baby Boomers—representing less than 5 years of Millennials' \$11,300 higher midcareer annual income. These higher educational investment costs for Millennials are also further offset by the increasing generosity of financial aid. For example, student aid in the form of grants (excluding loans and work study programs) were on average \$26,000 higher for Millennials than Generation X, cutting the college cost differential by two thirds such that it would take only 3 years for Millennials to make up the net college cost difference.

Despite the income growth from one generation to the next that we observe, there are also areas in which Millennials fell behind previous generations—in particular their homeownership rates. In 2022 (when the oldest Millennials were age 41) 61 percent of people ages 35–44 owned a home, down from 66 percent in 1989 (when the oldest Baby Boomers were age 43) (Federal Reserve Board 2023b). This is consistent with rising housing prices making it more difficult for young adults to purchase their first home. Yet, at the same time, the share of 35–44 year olds who hold stocks reached 64 percent in 2022, up from 39 percent in 1989 (Federal Reserve Board 2023b). Additionally, when looking at overall wealth holdings, Horpedahl (2021) found intergenerational wealth growth that is consistent with our increases in intergenerational incomes. Hence, while the types of asset holdings have shifted across generations, this shift does not appear to have prevented broader growth in wealth from one generation to the next.

Aside from comparing our results to popular perceptions of intergenerational progress, it is informative to compare our results to related research, especially Chetty et al. (2017) who find that rates of absolute mobility have slowed over time. They calculate in each year the share of adults aged 30 whose income exceeds that of their parents when their parents were around age 30. Whereas around 90 percent of 30-year olds born in the early 1940s had higher incomes than their parents at the same age, this was true for just over half of 30-year olds born in the early 1980s. This decline in absolute mobility is consistent with our finding of slowing intergenerational progress from the Silent Generation (born 1928–1945) through Generation X (born 1965–1980). But our results suggest that intergenerational progress is no longer slowing

24

and may instead be picking up again for Millennials in their late 30s. While Chetty et al. (2017) report trends in absolute mobility for 40-year old adults as a sensitivity check, they are only able to examine 40-year olds born through the early 1970s, several years before Millennials were born. That said, it is not necessarily the case that absolute mobility will move in tandem with outward shifts in the distribution of income over generations. Notably, Chetty et al. (2017) find that simply increasing economic growth without changing the proportion of resources allocated to households would do less to increase absolute mobility than allocating a greater share of existing resources to lower income households. But it remains possible that with more years of data absolute mobility trends will stop falling and begin to tick up, consistent with the intergenerational trends we find.

One caveat for our results is that we rely on survey data affected by misreporting of income. Underreporting has been documented for earnings (e.g., Bollinger et al. 2019), taxes (e.g., Meyer et al. 2020), and transfer programs (e.g., Meyer, Mok, and Sullivan 2015). While our focus on the median largely avoids the most pervasive problems with the growing underreporting of means tested transfers (Meyer, Mok, and Sullivan 2015), misreported earnings and other income sources along with inaccurate tax calculations could still bias our results. For example, Corinth, Meyer, and Wu (2022) find that between 1995 and 2016, the share of aggregate earnings reported in the CPS ASEC fell by 3 percentage points, the share of Unemployment Insurance fell by 27 percentage points, and the share of Earned Income Tax Credit dollars fell by 7 percentage points. Worsening underreporting of income would bias our estimates of intergenerational progress downward, especially for more recent generations. Thus, the intergenerational progress of Millennials likely compares more favorably to the intergenerational progress of preceding generations than we report.

Another caveat is that our conclusions include three years of data around the COVID-19 pandemic. The historic rise in unemployment in 2020 led to severe reductions in earnings for a substantial share of workers, while government policies including expanded and more generous Unemployment Insurance, three rounds of Economic Impact Payments (e.g., stimulus payments), and a temporary expansion of the Child Tax Credit increased post-tax incomes—which substantially reduced poverty rates over that period (Shrider and Creamer 2023). While many COVID-related government benefits ended by 2022, the pandemic recovery was also notably

25

progressive in its distribution (Larrimore, Mortenson, and Splinter 2023b). These recent years of data do not appear to drive our results—we conducted similar analyses using data only through 2019 and found similar patterns. Nevertheless, it is possible that generational income patterns in the past several years may differ from that which we should expect in the future and it is possible that intergenerational progress as the Millennial generation ages increases or declines from that seen so far.

### 9. Conclusion

Using data from 1963 through 2022, we evaluate whether younger generations are seeing slower income growth relative to the generations that came before. We confirm that there has been a slowdown in intergenerational progress, except for Millennials who saw their incomes grow slightly faster than Generation X but still more slowly than Baby Boomers and the Silent Generation. Intergenerational progress has remained positive for all generations. Positive growth has been maintained for Generation X and Millennials in spite of their stalled growth in hours worked.

We investigate the role of two potential explanations for perceptions of worsening outcomes for Millennials despite their observed income growth relative to previous generations. First, we find that the higher household incomes of Millennials relative to Generation X, through their 20s, is a result of dependence on their parents rather than a rise in their own market incomes. By age 31, however, less than 10 percent of Millennials are still dependent on their parents and by then their own market incomes exceed that of previous generations. Second, we find that the rising cost of college offsets only a small portion of the income gains achieved by Millennials, especially when accounting for the growing generosity of financial aid.

Our results focus on aggregate comparisons across generations, as opposed to direct comparisons between individuals and their own parents. Each type of comparison provides important information about absolute improvements in economic wellbeing across generations. Future research should continue to consider alternative measures of wellbeing for evaluating intergenerational progress, including consumption, wealth and social wellbeing (e.g., Fisher and Johnson 2022). Results on changes in wellbeing over time, including the intergenerational progress made in rising incomes, should inform discussions about how best to promote wellbeing in the future.

### References

- Armour, Philip, Richard V Burkhauser, and Jeff Larrimore. 2013. "Deconstructing Income and Income Inequality Measures: A Crosswalk from Market Income to Comprehensive Income." *American Economic Review* 103 (3): 173–77. https://doi.org/10.1257/aer.103.3.173.
- Auten, Gerald, and David Splinter. Forthcoming. "Income Inequality in the United States: Using Tax Data to Measure Long-Term Trends." *Journal of Political Economy*. https://doi.org/10.1086/728741.
- Board of Governors of the Federal Reserve System. 2023. "Changes in U.S. Family Finances from 2019 to 2022: Evidence from the Survey of Consumer Finances." *Federal Reserve Bulletin*, October. https://doi.org/10.17016/8799.
- Bollinger, Christopher R., Barry T. Hirsch, Charles M. Hokayem, and James P. Ziliak. 2019.
  "Trouble in the Tails? What We Know about Earnings Nonresponse 30 Years after Lillard, Smith, and Welch." *Journal of Political Economy* 127 (5): 2143–85. https://doi.org/10.1086/701807.
- Brenan, Megan. 2022. "Americans Less Optimistic About Next Generation's Future." *Gallup*, October 25, 2022. https://news.gallup.com/poll/403760/americans-less-optimistic-next-generation-future.aspx.
- Burkhauser, Richard V., Kevin Corinth, James Elwell, and Jeff Larrimore. Forthcoming. "Evaluating the Success of the War on Poverty Since 1963 Using an Absolute Full-Income Poverty Measure." *Journal of Political Economy*, 725705. https://doi.org/10.1086/725705.
- Burkhauser, Richard V., Jeff Larrimore, and Kosali I. Simon. 2012. "A 'Second Opinion' on the Economic Health of the American Middle Class." *National Tax Journal* 65 (1): 7–32. https://doi.org/10.17310/ntj.2012.1.01.
- Canilang, Sara, Cassandra Duchan, Kimberly Kreiss, Jeff Larrimore, Ellen Merry, Erin Troland, and Mike Zabek. 2020. "Report on the Economic Well-Being of U.S. Households in 2019, Featuring Supplemental Data from April 2020." Board of Governors of the Federal Reserve System. https://www.federalreserve.gov/publications/files/2019-reporteconomic-well-being-us-households-202005.pdf.
- Census Bureau. 2022. "Historical Households Tables." https://www.census.gov/data/tables/time-series/demo/families/households.html.
- Centers for Medicare and Medicaid Services. 2023. "National Health Expenditure Accounts." https://www.cms.gov/data-research/statistics-trends-and-reports/national-healthexpenditure-data/historical.
- Chetty, Raj, David Grusky, Maximilian Hell, Nathaniel Hendren, Robert Manduca, and Jimmy Narang. 2017. "The Fading American Dream: Trends in Absolute Income Mobility since 1940." *Science* 356 (6336): 398–406. https://doi.org/10.1126/science.aal4617.
- Chetty, Raj, Nathaniel Hendren, Patrick Kline, and Emmanuel Saez. 2014. "Where Is the Land of Opportunity? The Geography of Intergenerational Mobility in the United States \*." *The Quarterly Journal of Economics* 129 (4): 1553–1623. https://doi.org/10.1093/qje/qju022.
- Congressional Budget Office. 2012. "The Distribution of Household Income and Federal Taxes, 2008 and 2009." Congress of the United States, Congressional Budget Office. https://www.cbo.gov/sites/default/files/114th-congress-2015-2016/reports/51361householdincomefedtaxes.pdf.

-. 2023. "The Budget and Economic Outlook: 2023 to 2033."

https://www.cbo.gov/system/files/2023-02/58848-Outlook.pdf.

- Corinth, Kevin, Bruce D. Meyer, and Derek Wu. 2022. "The Change in Poverty from 1995 to 2016 among Single-Parent Families." *AEA Papers and Proceedings* 112 (May): 345–50. https://doi.org/10.1257/pandp.20221043.
- Dimock, Michael. 2019. "Defining Generations: Where Millennials End and Generation Z Begins." *Pew Research Center*, January 17, 2019. https://www.pewresearch.org/shortreads/2019/01/17/where-millennials-end-and-generation-z-begins/.
- Dynarski, Susan, Lindsay Page, and Judith Scott-Clayton. 2022. "College Costs, Financial Aid, and Student Decisions." w30275. Cambridge, MA: National Bureau of Economic Research. https://doi.org/10.3386/w30275.
- Elwell, James, Kevin Corinth, and Richard Burkhauser. 2020. "Income Growth and Its Distribution from Eisenhower to Obama: The Growing Importance of In-Kind Transfers (1959-2016)." In United States Income, Wealth, Consumption, and Inequality. Oxford University Press.
- Federal Reserve Board. 2023a. "Survey of Household Economics and Decisionmaking." https://www.federalreserve.gov/consumerscommunities/shed.htm.
- 2023b. "Survey of Consumer Finances (SCF), 1989 2022."
   https://www.federalreserve.gov/econres/scf/dataviz/scf/chart/#series:Primary\_Residence;
   demographic:agecl;population:1,2;units:median;range:1989,2022.
- Feenberg, Daniel, and Elisabeth Coutts. 1993. "An Introduction to the TAXSIM Model." Journal of Policy Analysis and Management 12 (1): 189. https://doi.org/10.2307/3325474.
- Filipovic, Jill. 2020. *OK Boomer, Let's Talk: How My Generation Got Left Behind*. New York, NY: One Signal Publishers.
- Fisher, Jonathan D., and Davis S. Johnson. 2022. "Inequality and Mobility over the Past Half-Century Using Income, Consumption, and Wealth." In *Measuring Distribution and Mobility of Income and Wealth*.
- Flood, Sarah, Miriam King, Renae Rodgers, Steven Ruggles, Warren J. Robert, and Michael Westberry. 2022. "Integrated Public Use Microdata Series, Current Population Survey: Version 10.0 [Dataset]." Minneapolis, MN: IPUMS. https://doi.org/10.18128/D030.V10.0.
- Guzman, Gloria, and Melissa Kollar. 2023. "Income in the United States: 2022." United States Census Bureau.

https://www.census.gov/content/dam/Census/library/publications/2023/demo/p60-279.pdf.

- Horpedahl, Jeremy. 2021. "Who Is the Wealthiest Generation?" *Economist Writing Every Day* (blog). September 1, 2021. https://economistwritingeveryday.com/2021/09/01/who-is-the-wealthiest-generation/.
- Julian, Christopher A. 2021. "Median Age at First Marriage, 2021." FP-22-15. Family Profiles. Bowling Green, OH: National Center for Family & Marriage Research. https://www.bgsu.edu/ncfmr/resources/data/family-profiles/julian-median-age-firstmarriage-2021-fp-22-15.html.
- Kaestner, Robert, and Darren Lubotsky. 2016. "Health Insurance and Income Inequality." *Journal of Economic Perspectives* 30 (2): 53–78. https://doi.org/10.1257/jep.30.2.53.

- Larrimore, Jeff, Richard V. Burkhauser, Gerald Auten, and Philip Armour. 2021. "Recent Trends in US Income Distributions in Tax Record Data Using More Comprehensive Measures of Income Including Real Accrued Capital Gains." *Journal of Political Economy* 129 (5): 1319–60. https://doi.org/10.1086/713098.
- Larrimore, Jeff, Jacob Mortenson, and David Splinter. 2023a. "Unemployment Insurance in Survey and Administrative Data." *Journal of Policy Analysis and Management* 42 (2): 571–79. https://doi.org/10.1002/pam.22463.
  - 2023b. "Earnings Business Cycles: The Covid Recession, Recovery, and Policy Response." *Journal of Public Economics* 225 (September): 104983. https://doi.org/10.1016/j.jpubeco.2023.104983.
- Larrimore, Jeff, Jake Mortenson, and David Splinter. 2022. "Presence and Persistence of Poverty in US Tax Data." In *Measuring Distribution and Mobility of Income and Wealth*, 383– 409. University of Chicago Press.
- Lee, Chul-In, and Gary Solon. 2009. "Trends in Intergenerational Income Mobility." *Review of Economics and Statistics* 91 (4): 766–72. https://doi.org/10.1162/rest.91.4.766.
- Long, Jason, and Joseph Ferrie. 2013. "Intergenerational Occupational Mobility in Great Britain and the United States Since 1850." *American Economic Review* 103 (4): 1109–37. https://doi.org/10.1257/aer.103.4.1109.
- Luhby, Tami. 2020. "Many Millennials Are Worse off than Their Parents a First in American History." *CNN*, January 11, 2020. https://www.cnn.com/2020/01/11/politics/millennials-income-stalled-upward-mobility-us/index.html.
- Mathews, T.J., and Brady E. Hamilton. 2002. "Mean Age of Mother, 1970–2000." *National Vital Statistics Reports* 51 (1). https://www.cdc.gov/nchs/data/nvsr/nvsr51/nvsr51\_01.pdf.
- Mayer, Susan E., and Leonard M. Lopoo. 2005. "Has the Intergenerational Transmission of Economic Status Changed?" *Journal of Human Resources* XL (1): 169–85. https://doi.org/10.3368/jhr.XL.1.169.
- Meyer, Bruce D., Wallace K.C. Mok, and James X. Sullivan. 2015. "The Underreporting of Transfers in Household Surveys: Its Nature and Consequences." University of Chicago Working Paper. Harris School of Public Policy.
- Meyer, Bruce D., and James X. Sullivan. 2012. "Winning the War: Poverty from the Great Society to the Great Recession." *Brookings Papers on Economic Activity*. https://www.brookings.edu/wp-content/uploads/2012/09/2012b meyer.pdf.
- Meyer, Bruce D., Derek Wu, Grace Finley, Patrick Langetieg, Carla Medalia, Mark Payne, and Alan Plumley. 2020. "The Accuracy of Tax Imputations: Estimating Tax Liabilities and Credits Using Linked Survey and Administrative Data." w28229. Cambridge, MA: National Bureau of Economic Research. https://doi.org/10.3386/w28229.
- Moulton, Brent R. 2018. "The Measurement of Output, Prices, and Productivity: What's Changed Since the Boskin Commission?" Brookings Institution. https://www.brookings.edu/wp-content/uploads/2018/07/Moulton-report-v2.pdf.
- National Center for Education Statistics. 2022. "Digest of Education Statistics." https://nces.ed.gov/programs/digest/2022menu\_tables.asp.
- Piketty, Thomas, Emmanuel Saez, and Gabriel Zucman. 2018. "Distributional National Accounts: Methods and Estimates for the United States." *The Quarterly Journal of Economics* 133 (2): 553–609. https://doi.org/10.1093/qje/qjx043.
- Rothbaum, Jonathan, and Adam Bee. 2021. "Coronavirus Infects Surveys, Too: Survey Nonresponse Bias and the Coronavirus Pandemic," United States Census Bureau

Working Paper, . https://www.census.gov/content/dam/Census/library/working-papers/2020/demo/sehsd-wp2020-10.pdf.

Shrider, Emily A., and John Creamer. 2023. "Poverty in the United States: 2022." United States Census Bureau. https://www.census.gov/content/dam/Census/library/publications/2023/demo/p60-

https://www.census.gov/content/dam/Census/library/publications/2023/demo/p60-280.pdf.

- Song, Xi, Catherine G. Massey, Karen A. Rolf, Joseph P. Ferrie, Jonathan L. Rothbaum, and Yu Xie. 2020. "Long-Term Decline in Intergenerational Mobility in the United States since the 1850s." *Proceedings of the National Academy of Sciences* 117 (1): 251–58. https://doi.org/10.1073/pnas.1905094116.
- Stewart, Kenneth J., and Stephen B. Reed. 1999. "Consumer Price Index Research Series Using Current Methods, 1978-98." *Monthly Labor Review*, CPI Research Series, , 29–38.
- Strauss, William, and Neil Howe. 1991. *Generations: The History of America's Future, 1584 to 2069.* New York: Morrow.
- Twenge, Jean M. 2023. Generations: The Real Differences Between Gen Z, Millennials, Gen X, Boomers, and Silents—and What They Mean for America's Future. New York: Atria Books.
- Urahn, Susan K., Erin Currier, Diana Elliott, Lauren Wechsler, Denise Wilson, and Daniel Colbert. 2012. "Pursuing the American Dream: Economic Mobility Across Generations." The Pew Charitable Trusts.

https://www.pewtrusts.org/~/media/legacy/uploadedfiles/wwwpewtrustsorg/reports/econ omic\_mobility/pursuingamericandreampdf.pdf.

- U.S. Bureau of Labor Statistics. 2019. "Frequently Asked Questions about the Chained Consumer Price Index for All Urban Consumers (C-CPI-U)." https://www.bls.gov/cpi/additional-resources/chained-cpi-questions-and-answers.htm.
- U.S. Department of Agriculture. 2023. "Budget Share for Total Food Increased 13 Percent in 2022." Economic Research Service. https://www.ers.usda.gov/data-products/chart-gallery/gallery/chart-detail/?chartId=76967.
- Van Dam, Andrew. 2020. "The Unluckiest Generation in U.S. History." *The Washington Post*, June 5, 2020. https://www.washingtonpost.com/business/2020/05/27/millennialrecession-covid/.
- Ward, Zachary. 2023. "Intergenerational Mobility in American History: Accounting for Race and Measurement Error." *American Economic Review* 113 (12): 3213–48. https://doi.org/10.1257/aer.20200292.

### **Figures and Tables**

Figure 1. Calendar years in which generation members of given age are observed, age 0 to 80



Note: Dashed red lines indicate the lower bound (1963) and upper bound (2022) calendar years in our sample. Generations are defined according to Pew Research. The vertical dashed black lines at ages 36 and 40 indicate our focal age range that spans five generations, the Greatest Generation through Millennials, in our observed sample. Source: Dimock (2019); Authors' Calculations



#### Figure 2. Median market income by age and generation, by sharing unit





Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Lost Generation (1883-1900), Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996), Generation Z (1997-2012). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

# Figure 3. Median post-tax, post-transfer income excluding health insurance by age and generation, by sharing unit





Figure 3b. Household sharing unit

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Lost Generation (1883-1900), Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996), Generation Z (1997-2012). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.



Figure 4. Labor force participation rate among adults aged 25-54, and mean when each generation aged 36-40, 1963-2022

Source: U.S. Bureau of Labor Statistics, retrieved from FRED, Federal Reserve Bank of St. Louis; Authors' calculations Notes: Red line indicates the prime age (25-54) labor force participation in the United States in each year. Dashed black lines span the period when a given generation's members were aged 36-40, and their level indicates the mean prime age labor force participation during those years. Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996).



20,000

10,000





Figure 5c. Bachelor's Degree

20,000

10,000

Figure 5d. Graduate Degree

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Lost Generation (1883-1900), Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

# Figure 6. Median post-tax, post-transfer income excluding health insurance by age and generation, by race and sharing unit



Figure 6a. Black adults, couple sharing unit



Figure 6b. Black adults, household sharing unit







Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Lost Generation (1883-1900), Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996), Generation Z (1997-2012). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.





Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Household income refers to total income received by all members of household, divided by the square root of the household size. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index. Solid lines indicate the older members of each generation (those born prior to or on the median year of the generation's birth year range), and dashed lines indicate the younger members of each generation (those born after the median year of the generation's birth year range).

# Figure 8. Median post-tax, post-transfer income including health insurance by age and generation, by sharing unit



Figure 8a. Couple sharing unit

Figure 8b. Household sharing unit

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2020; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and includes the market value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.



# Figure 9. Median income excluding health insurance by age and generation using CPI-U-RS inflation

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Lost Generation (1883-1900), Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996), Generation Z (1997-2012). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and includes the market value of health insurance. All income values are converted to real dollars using the Consumer Price Index retroactive series.

	Greatest	Silent	Boomer	Gen-X	Millennials	
Panel A. Market income						
Couple		34.7	23.0	8.0	11.1	
Household	—	43.3	31.1	9.8	13.7	
Panel B. Post-tax, post-transfer income (excluding health insurance)						
Couple		25.1	18.8	15.0	15.7	

Household

### Table 1. Percent increase in median income from previous generation at ages 36-40

34.0 Source: Current Population Survey Annual Social and Economic Supplement, 1964-2023; Burkhauser et al. (Forthcoming); Authors' calculations

26.8

16.1

18.4

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

	Greatest	Silent	Boomer	Gen-X	Millennials
Panel A. Market Income					
Couple	—	27.8	10.3	-1.2	11.4
Household		39.3	20.5	0.9	12.6

### Table 2. Percent increase in 25<sup>th</sup> percentile income from previous generation at ages 36-40

Panel B. Post-tax, post-transfer income (excluding health insurance) 23.7 10.0

Couple

32.3 20.5 11.7 19.8 Household Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

10.5

16.5

### Table 3. Percent increase in 75<sup>th</sup> percentile income from previous generation at ages 36-40

	Greatest	Silent	Boomer	Gen-X	Millennials
Panel A. Market income					
Couple	—	38.9	28.9	15.8	14.0
Household	—	47.1	35.8	17.2	16.7

#### Panel B. Post-tax, post-transfer income (excluding health insurance)

Couple	, i i i i i i i i i i i i i i i i i i i	 23.9	23.2	21.4	16.5
Household		 33.2	30.1	21.5	20.2

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

	Greatest	Silent	Boomer	Gen-X	Millennials	
Panel A. Hours per year						
Median	1,116	1,414	1,874	1,894	1,968	
Mean	1,298	1,474	1,685	1,654	1,655	

# Table 4. Hours of paid work per year by individual and spouse, by generation at ages 36–40

Panel B. Percent change in hours per year from previous generation

Median	 27%	32%	1%	4%
Mean	 14%	14%	-2%	0%

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Authors' calculations Notes: Hours represent the total number of paid hours worked per year by the member of the generation aged 36-40 and their spouse, if any. Inclusion in the sample is not conditioned on participation in work. Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965– 1980), Millennials (1981–1996).

# Table 5. Percent change in market income under various work hour scenarios for the mean of the middle quintile of the market income distribution, couple sharing unit, by generation at ages 36–40

	Greatest	Silent	Boomer	Gen-X	Millennials
Actual		32.3%	21.6%	8.1%	10.8%
Holding work hours constant		14.0%	2.2%	6.8%	10.0%
Due to work hours change	_	18.2%	19.4%	1.3%	0.7%

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Authors' calculations Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index. Income values represent the mean value within the middle quintile of the market income distribution. Market income holding work hours constant counterfactually assumes the younger generation worked the same mean number of hours as the older generation in each pairwise comparison of generations. Market income due to work hours change is the difference between the actual market income growth rate and market income growth rate holding work hours constant.

	High-school degree or less	Some college / Assoc. degree	Bachelor's degree	Graduate degree
Greatest	76.2	10.2	8.7	4.9
Lost	68.0	14.1	9.7	8.2
Boomer	47.3	25.8	17.3	9.7
Gen-X	38.1	26.4	22.6	12.9
Millennial	32.5	24.7	26.1	16.6

Table 6. Educational attainment shares by generation at ages 36-40

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Authors' calculations Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Values represent the share of each generation's members who had a given level of educational attainment, pooling across all members aged 36-40 including each year in which each individual was in this age range.

	Greatest	Silent	Boomer	Gen-X	Millennials
Couple sharing unit					
High-school or less		22.1	5.2	1.8	12.2
Some College or Assoc. degree		18.6	10.1	7.1	9.0
Bachelor's degree		19.0	16.9	16.8	9.9
Graduate degree		21.4	20.4	24.5	12.1
Household sharing unit					
High-school or less		30.8	14.1	3.8	17.1
Some College or Assoc. degree		29.0	16.7	7.0	12.9
Bachelor's degree		25.6	24.5	17.8	12.8
Graduate degree		30.3	27.4	23.7	15.4

## Table 7. Percent increase in median post-tax, post-transfer income from previous generation at ages 36–40, by educational attainment

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

# Table 8. Average tuition, fees, room, and board in the year that the median aged individual of each cohort turned 21

	Tuition, fees, room and board
Baby Boomer (1975-1976)	\$7,824
Generation-X (1988-1989)	\$10,811
Millennial (2009-2010)	\$20,217

Source: National Center for Education Statistics (2022)

Notes: Academic years shown represent the middle year in which members of the given generation turned age 21. Values represent the total cost of tuition, fees, room and board in a given academic year, before financial aid is applied. Values are converted in 2019 dollars using the Personal Consumption Expenditure Price Index.

	Greatest	Silent	Boomer	Gen-X	Millennials
Panel A. Black					
Couple	_	52.1	24.8	20.8	14.3
Household		68.8	36.0	19.4	20.2
Panel B. Nonblack					
Couple	—	23.8	18.9	15.0	15.8
Household		33.0	27.2	16.4	18.0

# Table 9. Percent increase in median post-tax post-transfer income from previous generation at ages 36-40, by race

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

	Greatest	Silent	Boomer	Gen-X	Millennials			
Panel A. Market income								
Couple			39.2	10.7	10.7			
Household			51.4	13.7	13.9			
Panel B. Post-tax, post-transfer income (excluding health insurance)								
Couple		_	23.3	21.3	16.7			

Household

# Table 10. Percent increase in median income from previous generation at ages 36–40 among the oldest 3 years in each generation only

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

35.2

22.6

20.2

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Only individuals born in the 3 final years of each generation are included. Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

Table 11. Percent increase in median post-tax, post-transfer income
from previous generation at ages 36–38, with and without health
insurance

	Greatest	Silent	Boomer	Gen-X	Millennial			
Panel A. Post-tax, post-transfer income (excluding health insurance)								
Couple		26.4	17.6	15.7	13.5			
Household	—	34.4	27.6	17.1	16.3			
Panel R Post-tax post-transfer income (including health insurance)								
Couple		28.1	22.3	19.1	13.3			
Household		36.2	33.6	22.2	15.7			

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2020; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and includes the market value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

	Greatest	Silent	Boomer	Gen-X	Millennials
Panel A. Market income using CH	PI-U-RS				
Couple	_	30.8	19.7	1.8	7.6
Household	—	38.9	27.7	3.5	10.2
Panel B. Post-tax, post-transfer in	ncome (exclud	ing health	insurance) u.	sing <b>CPI-U</b> -	-RS
Couple	—	21.1	15.8	8.5	12.0
Household	—	29.9	23.5	9.6	14.5
Panel C. Market income using <b>Pa</b>	CE				
Couple	_	34.7	23.0	8.0	11.1
Household	—	43.3	31.1	9.8	13.7
Panel D. Post-tax, post-transfer i	ncome (exclud	ling health	insurance) u	sing <b>PCE</b>	
Couple	_	25.1	18.8	15.0	15.7
Household	_	34.0	26.8	16.1	18.4

# Table 12. Percent increase in median income from previous generation at ages 36–40 using alternative measures of inflation

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

### **Appendix A. Appendix Figures and Tables**

Figure A1. Share of Generation's Members Dependent on Their Parents or Spouse's Parents, by Age



Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996), Generation Z (1997–2012). An individual is defined as dependent if he or she lives in a household in which his or her parents, or his or her spouse's parents, receive more than 50 percent of the household's post-tax, post-transfer income. Post-tax, post transfer income includes market income and all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance.



Figure A2. Mean Work Hours by Age and Generation

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Authors' calculations Notes: Generations are defined by the following birth years: Lost Generation (1883-1900), Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996), Generation Z (1997-2012). Work hours for married couples refers to total work hours of by both members of the married couple divided by two, and for unmarried individuals refers to total work hours by the individual. Work hours are calculated based on responses to the number of hours of work per week and number of weeks worked in the past year. Inclusion in the sample is not conditioned on participation in work.

	Greatest	Silent	Boomer	Gen-X	Millennials	
Panel A. Market Income						
Couple		33.2	-14.6	-11.5	7.5	
Household		40.4	6.8	-2.1	13.7	

# Table A1. Percent increase in 10th percentile income from previousgeneration at ages 36–40

Panel B. Post-tax, post-transfer income (excluding health insurance)

Couple	 28.4	2.6	6.6	17.4
Household	 42.2	17.6	15.9	20.7

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

	Greatest	Silent	Boomer	Gen-X	Millennials
Panel A. Market income					
Couple		42.1	33.9	22.5	16.9
Household	—	50.5	40.6	22.8	18.6
Panal R Post tax post transform	incomo (orchudi	na haalth i	nsurance)		
r unei D. r osi-iux, posi-iransjer i	encome (exclual	ng neulin i	nsurance)		
Couple		22.7	28.9	25.9	18.6

# Table A2. Percent increase in 90th percentile income from previousgeneration at ages 36–40

Household

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

31.0

34.8

26.0

21.6

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Market income includes all sources of income not received from government sources, does not adjust for taxes, and excludes the value of health insurance. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

	Greatest	Silent	Boomer	Gen-X	Millennials	
Panel A. Implied hourly wage (in 2019 dollars)						
Median	17.35	20.38	21.17	23.31	25.74	
Mean	21.76	25.06	25.95	31.09	34.60	

# Table A3. Implied hourly wage for individual and spouse, among workers by generation at ages 36–40

Panel B. Percent change in implied hourly wage from previous generation

Median	—	18%	4%	10%	10%
Mean		15%	4%	20%	11%

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Authors' calculations Notes: Hourly wage represents the total labor earnings for the individual and spouse, if any, divided by the total number of paid hours worked per year by the member of the generation aged 36-40 and their spouse, if any. Hourly wages are expressed in real 2019 dollars, using the Personal Consumption Expenditures price index to adjust for inflation. Inclusion in the sample requires that either the individual or their spouse worked. Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981– 1996).

	Greatest	Silent	Boomer	Gen-X	Millennials
Panel A. Male					
Couple	_	24.1	16.8	13.6	15.0
Household	—	34.9	26.6	16.9	17.4
Panel B. Female					
Couple		26.7	20.7	16.4	16.4
Household		33.9	27.1	14.8	19.9

# Table A4. Percent increase in median post-tax post-transfer income from previous generation at ages 36-40, by sex

Source: Current Population Survey Annual Social and Economic Supplement, 1964–2023; Burkhauser et al. (Forthcoming); Authors' calculations

Notes: Generations are defined by the following birth years: Greatest Generation (1901–1927), Silent Generation (1928–1945), Baby Boomers (1946–1964), Generation X (1965–1980), Millennials (1981–1996). Couple income for married couples refers to total income received by both members of the married couple divided by two, and for unmarried individuals refers to total income received by the individual. Household income refers to total income received by all members of household, divided by the square root of the household size. Post-tax, post transfer income includes market income as well as all non-medical cash and in-kind transfers, adjusts for taxes, and excludes the value of health insurance. All income values are converted to real dollars using the Personal Consumption Expenditure price index.

### Appendix B. Approach for holding work hours constant across generations

We first show that the mean market income of a couple can be written as the product of (i) the weighted mean of the hourly wage, using hours worked as the weights, and (ii) hours worked, plus (iii) mean non-labor income.

$$\begin{split} \overline{m_{t}} &= \frac{1}{N_{t}} \sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} m_{i,t} \\ &= \frac{1}{N_{t}} \sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} w_{i,t} h_{i,t} + n_{i,t} \\ &= \frac{1}{N_{t}} \sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} w_{i,t} h_{i,t} + \overline{n_{t}} \\ &= \frac{1}{N_{t}} \sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} w_{i,t} h_{i,t} \left( \frac{\sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} h_{i,t}}{\sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} h_{i,t}} \right) + \overline{n_{t}} \\ &= \left( \sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} \frac{h_{i,t}}{\sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} h_{i,t}} w_{i,t} \right) \frac{1}{N_{t}} \sum_{\substack{i=1 \\ i=1}}^{N_{t}} h_{i,t} + \overline{n_{t}} \\ &= \left( \sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} \frac{h_{i,t}}{\sum_{\substack{i=1 \\ N_{t}}}^{N_{t}} h_{i,t}} w_{i,t} \right) \overline{h_{t}} + \overline{n_{t}} \end{split}$$

where  $m_{i,t}$  is the market income,  $w_{i,t}$  is the hourly wage,  $h_{i,t}$  is the hours worked, and  $n_{i,t}$  is the non-labor income of individual *i* in generation *t*, and  $N_t$  is the number of couples in generation *t* (or in our case the number of couples in the middle quintile of the market income distribution).

We next show that replacing the mean hours of generation t with the mean hours of generation t - 1 in Equation (1) adjusts the hours of each couple proportionally, by the ratio of mean hours of generation t - 1 to mean hours of generation t.

$$\begin{aligned} Adjusted \ \overline{m_t} &= \left(\sum_{i=1}^{N_t} \frac{h_{i,t}}{\sum_{i=1}^N h_{i,t}} w_{i,t}\right) \overline{h_{t-1}} + \overline{n_t} \\ &= \left(\sum_{i=1}^{N_t} \frac{h_{i,t}}{\sum_{i=1}^N h_{i,t}} w_{i,t}\right) \overline{h_t} \frac{\overline{h_{t-1}}}{\overline{h_t}} + \overline{n_t} \\ &= \left(\sum_{i=1}^{N_t} \frac{h_{i,t}}{\sum_{i=1}^N h_{i,t}} w_{i,t}\right) \left(\frac{1}{N} \sum_{i=1}^{N_t} h_{i,t}\right) \frac{\overline{h_{t-1}}}{\overline{h_t}} + \overline{n_t} \\ &= \frac{1}{N} \sum_{i=1}^{N_t} \frac{\overline{h_{t-1}}}{\overline{h_t}} h_{i,t} w_{i,t} + n_{i,t} \end{aligned}$$